%%    TEMPLATE for articles submitted to the full-tex econf proceedings
%%     
%%
%%     Please do not remove lines commented out with %+
%%           these are for the editors' use.
%%
%%     Questions?  Send email to :  mpeskin@slac.stanford.edu
%%   

\documentclass[12pt]{article}
\usepackage{epsfig}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%  Do not change these:
\textwidth=6.0in  \textheight=8.25in

%%  Adjust these for your printer:
\leftmargin=-0.3in   \topmargin=-0.20in


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%  personal abbreviations and macros
%    the following package contains macros used in this document:
\input econfmacros.tex
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%  To include an item in the INDEX of the conference volume,
%           flag it with    \index{<item name>}
%  The use of this macro is illustrated in the text.
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\def\Title#1{\begin{center} {\Large {\bf #1} } \end{center}}

\begin{document}
\begin{thebibliography}{99}

%%
%%  bibliographic items can be constructed using the LaTeX format in SPIRES:
%%    see    http://www.slac.stanford.edu/spires/hep/latex.html
%%  SPIRES will also supply the CITATION line information; please include it.
%%

% \bibitem{lumi}
% CMS Collaboration, {\it Absolute luminosity normalization}, CMS Detector Performance Summary DP-2011-002.

% \bibitem{fewz}
%   K.~Melnikov, F.~Petriello, Phys. Rev. {\bf D74}, (2006) 114017

% \bibitem{resbos}
%  C.~Balazs, C.P.~Yuan, {it Soft gluon effects on lepton pairs at hadron colliders},
%   Phys. Rev. {\bf D56} (1997) 5558

\bibitem{wz}
CMS Collaboration, {\it Measurements of the Inclusive W and Z Production Cross Sections in pp Collisions at $\sqrt{s}=7\,\mathrm{TeV}$}, Submitted to JHEP,
arXiv:1107.4789 (2011).

\bibitem{wasym} 
CMS Collaboration, {\it Measurement of the Lepton Charge Asymmetry in Inclusive W Production in pp Collisions at $\sqrt{s}=7\,\mathrm{TeV}$}, CMS Physics Analysis Summary EWK-10-006.

\bibitem{dy}
CMS Collaboration, {\it Measurement of the Drell-Yan Differential Cross Section in pp Collisions at $\sqrt{s}=7\,\mathrm{TeV}$}, CMS Physics Analysis Summary EWK-10-007.

\bibitem{zdiff}
CMS Collaboration, {\it Measurement of Differential Cross Sections for Z Bosons at $\sqrt{s}=7\,\mathrm{TeV}$}, CMS Physics Analysis Summary EWK-10-010.

\bibitem{fb}
CMS Collaboration, {\it Measurement of Forward-Backward Asymmetry of Lepton Pairs and the Weak mixing Angle at CMS}, CMS Physics Analysis Summary EWK-10-011.

\bibitem{vjets}
CMS Collaboration, {\it Rates of Jets Produced in Association with W and Z Bosons in pp Collisions at $\sqrt{s}=7\,\mathrm{TeV}$}, CMS Physics Analysis Summary EWK-10-012.

\bibitem{bg}
F.A. Berends, W.T. Giele, H. Kuijf {\it et al.}, {\it Multi-jet production in W, Z events at p and anti-p colliders''}, Phys. Lett. {\bf B224} (1989) 237.

\bibitem{ztau}
CMS Collaboration, {\it Measurement of the Inclusive Z Cross Section via Decays to Tau Pairs in pp Collisions at $\sqrt{s}=7\,\mathrm{TeV}$}, CMS Physics Analysis Summary EWK-10-013.

\bibitem{wpol}
CMS Collaboration, {\it Measurement of the Polarization of W Bosons with Large Transverse Momenta in W+Jets Events at the LHC}, Phys. Rev. Lett. {\bf 1207} (2011) 021802.

\bibitem{zb}
CMS Collaboration, {\it Observation of Z+b, Z$\rightarrow$ee, μμ with CMS at $\sqrt{s}=7\,\mathrm{TeV}$}, CMS Physics Analysis Summary EWK-10-015.

\bibitem{wtau}
CMS Collaboration, {\it Observation of W$\rightarrow\tau\nu$ Production in pp Collisions at $\sqrt{s}=7\,\mathrm{TeV}$}, CMS Physics Analysis Summary EWK-11-002.

\bibitem{vgamma}
CMS Collaboration, {\it Measurement of W$\gamma$γ and Z$\gamma$γ production in pp collisions at $\sqrt{s}=7\,\mathrm{TeV}$}, CMS Physics Analysis Summary EWK-10-008.

\bibitem{ww}
CMS Collaboration, {\it Measurement of W$^+$W$^−$ production and search for the Higgs boson in pp collisions at $\sqrt{s}=7\,\mathrm{TeV}$}, Phys. Lett. {\bf B699}, 25-47 (2011).

%\bibitem{isogamma}
%CMS Collaboration, {\it Measurement of the Isolated Prompt Photon Production Cross Section in pp Collisions at $\sqrt{s}=7\,\mathrm{TeV}$} Phys. Ref. Lett. {\bf 106}, 082001 (2011).

\bibitem{lepasym}
CMS Collaboration, {\it Measurement of the lepton charge asymmetry in inclusive W production in pp collisions at $\sqrt{s}=7\,\mathrm{TeV}$}, JHEP {\bf 04}, 050 (2011).

%\bibitem{wzpaper}
%CMS Collaboration, {\it Measurements of inclusive W and Z cross sections in pp collisions at $\sqrt{s}=7\,\mathrm{TeV}$}, JHEP {\bf 01}, 080 (2011)


\end{thebibliography}



\Title{Electroweak results from CMS}

\bigskip\bigskip

%+\addtocontents{toc}{{\it D. Reggiano}}
%+\label{ReggianoStart}

\begin{raggedright}  

{\it Luca Lista\index{Lista, L.}\\
INFN Sezione di Napoli\\
Comp. Univ. M.S.Angelo, via Cintia\\
I-80126 Napoli, ITALY}
\end{raggedright}

%  \abstract
%  We present the results of electroweak studies performed using data collected in 2010 at a center-of-mass energy 
%  of 7~TeV by the CMS experiment at the LHC. Besides their intrinsic interest as unique samples to calibrate and
%   understand the CMS detector response to leptons, jets and missing energy, events containing W and Z bosons appear 
%  as dominant components in many Higgs seaches and in most of the searches beyond the Standard Model, either as signal 
%  or as background. In addition, the excellent level of theoretical and experimental understanding of these processes 
%  allows electroweak tests at the LHC at an unprecendented level of precision. CMS uses a wide range of final states 
%  to measure cross sections, asymmetries, polarizations and differential distributions in general. The integrated 
%  luminosity in 2010 was already sufficient to perform not just inclusive measurements using W and Z decays into 
%  electrons and muons, but also precise studies of associated jet production and final states containing taus, 
%  as well as first measurements of diboson cross sections and associated b-jet production.

\section{Introduction}
% W and Z production in pp collisions proceeds mainly form the scattering of a valence quark with a sea anti-quark.
% The involved parton fractions are low ($10^{-3} < x < 10^{-1}$) and scattering of a sea quark with a sea anti-quark is also important,
% and W production is charge asymmetric: $\sigma(\mathrm{W}^+)/\sigma(\mathrm{W}^−)\sim 1.43$, different from two, as one would expect 
% considering the contributions from valence plus sea quark scattering only, in the standard model.
Events with W and Z bosons in high energy pp collisions at LHC produce very clean signals and allow 
to perform precision measurements. 
Accurate theoretical standard model (SM) predictions at next-to-next-to-leading order (NNLO) in perturbative QCD
for cross section and differential distributions are available
with programs like {\sc FEWZ}, {\sc RESBOS}, {\sc DYNNLO}, and next-to-leading order (NLO) programs 
like {\sc POWHEG} and {\sc MC@NLO} provide full event generation.
Uncertainties in valence and sea PDF limit the accuracy of theoretical predictions, and, on the other hand,
differential distributions of eletcroweak processes are sensitive to PDF and can improve the precision of their
current knowledge.
Many estimates required to perform measurements of electroweak processes can be done 
on the large control samples available in data instead of relying on simulations.
Events containing W and Z are themselves useful tools to understand and calibrate the detector,
and the tag-and-proble (T\&P) method is extensively used to determine lepton efficiencies from high-purity samples of $\mathrm{Z}\rightarrow\ell^+\ell^-$ events, which are also used to study lepton momentum scale and resolution.
Many searches for new physics have electroweak processes among their main backgrounds, hence 
studying electroweak processes is also important to keep backgrounds to such searches under control.

\section{Studies of W and Z production}

W events are selected using a loose single-lepton trigger~\cite{wz}, and
well understood lepton identification cuts are applied. Leptons are required to have
$p_{\mathrm T}>25$~GeV and pseudorapidity $\eta$ within the trigger fiducial volume.
Lepton isolation is enforced requiring a small sum of tracker and calorimeter activities
within a cone of $\Delta R=\sqrt{(\Delta\varphi)^2+(\Delta\eta)^2}<0.3$. %, normalized to the lepton $p_{\mathrm T}$.
Events with a second lepton are vetoed to suppress the Drell--Yan (DY) background.
The W signal yield is extracted from a fit to the missing $E_{\mathrm T}$ distribution of selected events,
where parameterized or fixed binned shapes are used to model signal and background distributions.
The shape of the QCD background is determined from data inverting lepton identification
or isolation selections.
Lepton efficiencies are determined as a function of $p_{\mathrm T}$ and $\eta$
using a T\&P method.
The misssing $E_{\mathrm T}$ distribution is studied using the dilepton recoil in Z events.
Separate fits to missing $E_{\mathrm T}$ spectra in events containing positive and negative leptons
allow to extract $\sigma(\mathrm{W}^+)$ and $\sigma(\mathrm{W}^-)$.
Alternatively, a simultaneous fit allow to to extract 
the inclusive cross section $\sigma(\mathrm{W}^\pm)$ and the ratio $\sigma(\mathrm{W}^+)/\sigma(\mathrm{W}^-)$, in which
several uncertainties cancel.

The selection of Z events~\cite{wz} proceeds requiring pairs of isolated leptons having
$p_{\mathrm T}>20$ (25~GeV) for Z$\rightarrow\mu^+\mu^-$ (Z$\rightarrow$e$^+$e$^-$) and $\eta$ within the 
fiducial detector and trigger acceptances. 
The invariant mass range $60 < m_{\ell\ell} < 120$~GeV is used for signal extraction, 
which is performed via a simultaneous fit of the Z signal yield and efficiencies 
using different dilepton categories for Z$\rightarrow\mu^+\mu^-$, or a simpler
cut-and-count analysis using an efficiency determination from a T\&P method
for Z$\rightarrow$e$^+$e$^-$.

Efficiencies as well as background and signal shapes are determined mainly from data, 
allowing to minimize the experimental systematic uncertainties. 
Theoretical uncertainties affect acceptance determinations. 
In particular, the PDF uncertainty, the modeling of initial-state radiation,
higher-order QCD effects (studied using the {\sc RESBOS} generator), electroweak corrections, 
final-state radiation (studied with {\sc HORACE}), and factorization and renormalization scale 
(estimated using {\sc FEWZ}).
The inclusive W and Z production cross sections at a center-of-mass energy of 7~TeV are determined as:

\vspace{-0.4cm}
{\footnotesize
\begin{equation}
\sigma(\mathrm{pp}\rightarrow\mathrm{W}X)\times{\cal B}(\mathrm{W}\rightarrow\ell\nu) 
= 10.30\pm 0.02 (\mathrm{stat.})\pm 0.10(\mathrm{syst.})\pm 0.10(\mathrm{th.})\pm 0.41 (\mathrm{lumi.})\,\mathrm{nb}\, ,
\end{equation}
\begin{equation}
\sigma(\mathrm{pp}\rightarrow\mathrm{Z}X)\times{\cal B}(\mathrm{Z}\rightarrow\ell^+\ell^-) 
= 0.974\pm 0.007 (\mathrm{stat.})\pm 0.007(\mathrm{syst.})\pm 0.018(\mathrm{th.})\pm 0.039(\mathrm{lumi.})\,\mathrm{nb}\, .
\end{equation}
}
\vspace{-0.4cm}

The study of Z$\rightarrow\tau^+\tau^-$~\cite{ztau} is a benchmark for searches using tau leptons,
like $\mathrm{H}^+\rightarrow\tau^+\nu$ and  $\mathrm{H}\rightarrow\tau^+\tau^-$. Tau candidates are
reconstructed using the particle flow algorithm that combines tracker and calorimeter measurements to determine 
particle candidates. The main uncertainty in the measurement comes from the knowledge of the tau identification efficiency,
which is determined directly from data.
The study of the decay W$\rightarrow\tau\nu$~\cite{wtau} is particularly challenging,
requiring a trigger based on the presence of a tau candidate associated to missing $E_{\mathrm T}$.
The results of W and Z inclusive cross sections measurements using decays to tau are in 
agreement with the standard-model (SM) predictions and with the measurements performed in the 
electron and muon channels.%, within their relatively limited precision.

In the ratios of W$^+$/W$^-$ and W/Z cross sections, luminosity and other uncertainties cancel.
The $\mathrm{W}^+/\mathrm{W}^−$ ratio is potentially sensitive to PDFs, while
the W/Z ratio provides the most stringent test of the SM.
The following inclusive production cross-section ratios have been measured:

\vspace{-0.4cm}
{\footnotesize
\begin{equation}
\frac{\sigma(\mathrm{pp}\rightarrow\mathrm{W}^+X)\times{\cal B}(\mathrm{W}^+\rightarrow\ell\nu)} 
{\sigma(\mathrm{pp}\rightarrow\mathrm{W}^-X)\times{\cal B}(\mathrm{W}^-\rightarrow\ell\nu)} 
= 1.421\pm 0.006 (\mathrm{stat.})\pm 0.014(\mathrm{syst.})\pm \pm 0.014 (\mathrm{th.})\,,
\end{equation}
\begin{equation}
\frac{\sigma(\mathrm{pp}\rightarrow\mathrm{W}X)\times{\cal B}(\mathrm{W}\rightarrow\ell\nu)} 
{\sigma(\mathrm{pp}\rightarrow\mathrm{Z}X)\times{\cal B}(\mathrm{Z}\rightarrow\ell^+\ell^-)} 
= 10.54\pm 0.07 (\mathrm{stat.})\pm 0.08(\mathrm{syst.})\pm \pm 0.16 (\mathrm{th.})\,.
\end{equation}
}
\vspace{-0.4cm}

Good overall agreement of the inclusive cross section measurements and their ratios with theoretical 
predictions at NNLO is found. 

The $\mathrm{W}^+/\mathrm{W}^−$ ratio is also measured as a function of the lepton pseudorapidity
$\eta$~\cite{wasym}, in order to achieve a larger sensitivity to PDFs, using a 
similar selection to the inclusive cross section analysis.
Two $p_{\mathrm T}$ thresholds (25, 30 GeV) are used to probe different phase-space regions.
The achieved precision is better than 1.6\% in most of the $\eta$ bins for both electron 
and muon channels, thanks to the small charge mis-identification, which is 0.1(barrel)-0.4(endcap)\% for electrons
and less than $10^{-4}$ for muons.
The large collected data samples allows to study the differential Z production cross sections 
as a function of the Z boson rapidity and transverse momentum~\cite{zdiff},
and the full DY cross section as a function of the lepton pair invariant mass~\cite{dy}.

Good agreement is found with theoretical NNLO predictions after an unfolding 
correction of both resolution and final-state radiation, not included in the theoretical
calculation. The study of the lepton forward-backward asymmetry as a function of 
the dilepton invariant mass allows to perform a measurement of the Weinberg angle:
\begin{equation}
\sin^2\theta_{\mathrm{eff}} = 0.2287\pm 0.0077(\mathrm{stat.})\pm 0.0036(\mathrm{syst.})\,.
\end{equation}

The study of W and Z production in association with hadronic jets is an
important test of perturbative NLO predictions and provides an estimate of one of the 
main backgrounds to searches for the Higgs boson and other new physics.
The ratios  $\sigma(V+n\,\mathrm{jets})/\sigma(V)$ of the prodiction cross section of W and Z 
associated with $n$ jets and the inclusive boson's production cross section are determined, 
considering jets with $E_{\mathrm T} > 30$~GeV~\cite{vjets}.
The measurements are dominated by systematic uncertainties, one of the main uncertainty
being the knowledge of the jet energy scale.
Agreement with {\sc MadGraph} is found while discrepancies with the {\sc PYTHIA} 
parton shower approximation are observed.
The Berends--Giele scaling~\cite{bg} has been tested measuring the ratios $\sigma(V+n\,\mathrm{jets})/\sigma(V+(n-1)\,\mathrm{jets})$,
confirming that the ratio is approximatly indepedent on the number of jets $n$.
The ratio as a function of $n$ has been parametrized as $\alpha + \beta\times n$, and the fitted values for
$\alpha$ and $\beta$, the latter close to zero, as expected, are compared with the simulation predictions.

Z production has been studied in association with at least one jets identified as 
produced by a b quark~\cite{zb}. A small missing $E_{\mathrm T}$ must accompany
the presence of the lepton pair from the Z decay, in order to suppress the background due to top-quark production. 
The purity of the b-tag selection is determined from data using a fit to the distribution of 
the invariant mass of tracks associated to the secondary vertex.
This study allows to probe the two main production mechanism: a b pair produced from quark-quark or 
gluon-gluon scattering (“fixed flavour”), or a single b quark produced at partonic level (“variable flavour”).
The ratio of Z production cross section in association with a b jet and 
in association with a generic jet has been fonund to be in agreement with NLO theoretical predictions:
 \begin{equation}
 \sigma(\mathrm{Z}+\mathrm{b})/
 \sigma(\mathrm{Z}+\mathrm{jet}) = 0.054\pm 0.016\, ,\,\mathrm{with}\,\,\mathrm{Z\rightarrow\mathrm{e}^+\mathrm{e}^-}\, ,
 \end{equation}
 \begin{equation}
 \sigma(\mathrm{Z}+\mathrm{b})/
 \sigma(\mathrm{Z}+\mathrm{jet}) = 0.046\pm 0.014\, ,\,\mathrm{with}\,\, \mathrm{Z\rightarrow\mu^+\mu^-}\, .
 \end{equation}
The study of W polarization in events with a W associated to hadronic jets~\cite{wpol}  
is a stringent test of detailed features of QCD prediction in a final state that is common
to searches for new physics that would present different distribution of the 
final-state lepton's kinematic variables.
The W polarization fractions $f_L$, $f_R$ and $f_0$ are determined from the distributions 
of the lepton projection on the reconstructed W momentum in the transverse plane 
for positive and negative leptons.
The differences of $L$ and $R$ polarization fraction for negative and positive W production are: 
$(f_L - f_R)^- = 0.240\pm 0.036 (\mathrm{stat.})\pm 0.031(\mathrm{syst.})$
and $(f_L - f_R)^+ = 0.310\pm 0.036 (\mathrm{stat.})\pm 0.017(\mathrm{syst.})$ for 
negatively and positively charged leptons respectively, and confirm the SM prediction of a
predominantly left-handed W production.



\section{Diboson production}

The study of Z and W production in association with a photon~\cite{vgamma}
probes final states common to new physics searches, as well as the triple gauge coupling
predictions in the SM. The estimate of the misidentified photon background
is very important for this analysis, and is performed mainly from data.
The measured cross sections agree with the SM predictions:
\begin{equation}
\sigma(\mathrm{pp}\rightarrow\mathrm{W}\gamma+X) 
\times{\cal B}(\mathrm{W}\rightarrow\ell\nu)
=56.3\pm5.0(\mathrm{stat.})\pm 5.0(\mathrm{syst.})\pm 2.3(\mathrm{lumi.})\,,
\end{equation}
\begin{equation}
\sigma(\mathrm{pp}\rightarrow\mathrm{Z}\gamma+X) 
\times{\cal B}(\mathrm{Z}\rightarrow\ell\ell)
=9.4\pm5.0(\mathrm{stat.})\pm 1.0(\mathrm{syst.})\pm 0.6(\mathrm{lumi.})\,.
\end{equation}
First limits are set on anomalous trilinear WW$\gamma$, ZZ$\gamma$ and Z$\gamma\gamma$ gauge
couplings at $\sqrt{s}$=7~TeV.
The study of WW production~\cite{ww} is a challenging task, and a benchmark for the search for
the Higgs boson with the decay $\mathrm{H}\rightarrow\mathrm{W^+W^-}$.
Using W decays to electrons and muons, part of the $\mathrm{W}\rightarrow\tau\nu$ contribution
is also included in the selected sample. The selection is based on a DY veto,
requiring the presence of a missing $E_{\mathrm T}$,
and removing same-flavour dilepton events with an invariant mass close to the Z mass.
$\mathrm{Z}\rightarrow\tau^+\tau^-$ suppression is achieved requiring the
missing $E_{\mathrm T}$ projection transverse to closest leptons to be greater than 
35~GeV, and events containing a top quark are vetoed using the number of jets present in 
the event and applying soft muon and b-tagging vetos.
13 events selected against a background estimate of $3.3 \pm 1.2$, and the WW cross section
is measured as:
\begin{equation}
	\sigma(\mathrm{WW}) = 41.1 \pm 15.3(\mathrm{stat.}) \pm 5.8(\mathrm{syst.}) \pm 4.5 (\mathrm{lumi.})\,.
\end{equation}
Limits to anomalous WW$\gamma$ and WWZ couplings are also set from this study.
Spin correlation determines the distribution of the azimuthal angular separation of 
the two leptons, which  is used to discriminate a possible
signal due to a Higgs boson decay to WW, resulting in limits to the production of the Higgs boson 
in the presence of a fourth generation.

\begin{figure}[htb]
\begin{center}
\epsfig{file=cmsth.eps,width=4.0in}
\caption{Ratio of CMS electroweak measurements to SM theoretical predictions.}
\label{fig:res}
\end{center}
\end{figure}

\section{Summary}

CMS produced many EWK measurements with the first 36 pb$^{-1}$ of LHC data at 7~TeV.
Precision measurements of inclusive W and Z production cross sections with large statistics
provide a test of the SM at the new energy scale explored by the LHC.
W and Z production associated to jets, including Z plus b-jets, is studied, and W polarization 
is been measured in events containing a W plus jets.
Detailed studies of differential cross sections and many observables, like asymmetries,
have been reported, as well as studies of diboson production: W$\gamma$, Z$\gamma$, WW.
All measurements are so far in agreement with theoretical predictions from the SM.
A summary of the ratios of CMS measurement to the SM predictions is shown in Fig.~\ref{fig:res}.
While many measurements are already limited by systematic uncertainties, the increase of
integrated luminosity will benefit many other measurements and will allow to access more rare
SM processes.




 
\end{document}

